

2024, Vol. 39, No. 7, 781-791 https://doi.org/10.1037/pag0000838

Reduced Interhemispheric Transfer in Older Adults: Evidence From a Divided Visual Field One-Back Task

Jean-François Delvenne

School of Psychology, University of Leeds

One of the pivotal structural changes observed in the ageing brain pertains to the corpus callosum, the largest neural pathway interconnecting the two cerebral hemispheres. Studies have highlighted the degeneration of the corpus callosum, particularly in its anterior segments, as individuals age. This prompts an essential question regarding the potential functional repercussions of these structural changes on interhemispheric communication among older adults. Two experiments were conducted to explore potential compromises in the interhemispheric transfer of visual working memory (VWM) in older adults. Both young individuals (aged 18–28 years) and healthy older adults (aged 65–85 years) engaged in modified versions of the one-back paradigm. In this task, stimuli were sequentially presented in either the left or right hemifield, and participants indicated whether each stimulus matched the preceding one. Notably, when two stimuli are matched, they could appear either in the same hemifield or in opposite hemifields. The results revealed that, in comparison to young adults, older adults demonstrated a significant increase in matching errors when the two stimuli were presented in opposite hemifields rather than the same hemifield. This new finding strongly suggests a reduced interhemispheric transfer of VWM in older adults, potentially attributed to age-related atrophy in the anterior part of the corpus callosum.

Public Significance Statement

Younger and older adults performed a memory task where pictures of abstract objects were presented on the left or right side of the display. Older adults commit more memory errors when the objects switch from one side of the display to the other, while this effect was not observed in younger adults. These findings indicate that ageing may be associated with a decline in the ability to transfer information between the two hemispheres of the brain, possibly resulting from deterioration of the corpus callosum.

Keywords: visual working memory, cognitive aging, older adults, interhemispheric transfer

Typical ageing has consistently been linked to cognitive decline across various domains, such as working memory, information processing speed, and executive cognitive function (Murman, 2015). This decline poses a significant challenge to our increasingly ageing societies. Yet, it remains unclear whether this decline in cognitive performance stems from a broad impact on the central nervous system or is rooted in localized structural changes. In that context, the age-related structural degradation of the corpus callosum (CC) emerges as a probable factor. This large white matter structure, which comprised more than 200 million myelinated axonal projections, establishes connections between corresponding cortical regions in the two cerebral hemispheres in an anterior–posterior topographical

Jean-François Delvenne D https://orcid.org/0000-0002-1990-9738

arrangement (Aboitiz et al., 1992; Gazzaniga, 2000). Numerous magnetic resonance imaging studies have consistently documented a reduction in both volume and microstructural integrity of the CC in old age (Davis et al., 2009; Delvenne et al., 2021; McLaughlin et al., 2007; Michielse et al., 2010; Ota et al., 2006; Sullivan et al., 2002, 2010). Hence, it is plausible to consider that the general deterioration of the CC could lead to a decline in the integration of sensory-motor and cognitive processes, ultimately resulting in widespread cognitive deterioration.

Although the precise mechanisms remain largely elusive, some evidence suggests a link between decreased cognitive performance and age-related changes in CC structure (Fling et al., 2011; Jokinen et

This article was published Online First July 18, 2024.

Robert West served as action editor.

The author thanks Michael Heenan, Cameron Overton, Claudie Salem, Myles Cartwright, Abbie Jones, Mia Newcombe, Eva Morris, Eloise Massey, and Alisha Juma for their assistance with the data collection. The findings appearing in this article have not been presented at peer-reviewed conferences. Data used in this study and other related materials have been made publicly available on the Open Science Framework and can be retrieved and accessed at https://osf.io/5qfpu/? view_only=9d026bf5daaa41a48739cd525fb890cb (Delvenne, 2024).

Open Access funding provided by the University of Leeds: This work is licensed under a Creative Commons Attribution 4.0 International License (CC BY 4.0; https://creativecommons.org/licenses/by/4.0). This license permits copying and redistributing the work in any medium or format, as well as adapting the material for any purpose, even commercially.

Jean-François Delvenne played a lead role in conceptualization, data curation, formal analysis, investigation, methodology, project administration, writing–original draft, and writing–review and editing.

Correspondence concerning this article should be addressed to Jean-François Delvenne, School of Psychology, University of Leeds, Leeds LS2 9JT, United Kingdom. Email: J.f.delvenne@leeds.ac.uk

al., 2007; Ryberg et al., 2011; Sullivan et al., 2010; Zahr et al., 2009). Even longitudinal data indicate a positive association between CC structural integrity and cognitive performance over time in the context of ageing (Sullivan et al., 2002). However, despite evidence linking the atrophy of the CC in healthy older individuals to general cognitive decline, the specific impact of an ageing CC on interhemispheric communication remains uncertain. Given the correlation between CC size and the quantity of myelinated and nonmyelinated transcallosal fibers required for transmitting neuronal impulses between hemispheric communication may experience some disruption in older age compared to younger adulthood.

Previous research investigating the impact of ageing on interhemispheric communication has primarily employed assessments of visuomotor interhemispheric transfer time (IHTT) through the utilization of the Poffenberger paradigm (Poffenberger, 1912). The paradigm examines variations in reaction times (RTs) by analyzing the interhemispheric (crossed, e.g., left hemisphere/right hand) and intrahemispheric (uncrossed, e.g., left hemisphere/left hand) response pathways, referred to as the crossed-uncrossed difference (CUD). Numerous studies have noted a lengthening of IHTT in older adults (Bellis & Wilber, 2001; Davis et al., 2012; Jeeves & Moes, 1996; Reuter-Lorenz & Stanczak, 2000), although some studies have failed to observe any ageing effects (Linnet & Roser, 2012). Evidence suggests that the CUD may rely on transfer through anterior CC regions, such as the CC midbody or genu (Gawryluk et al., 2011; Omura et al., 2004; Tettamanti et al., 2002; Weber et al., 2005). The potential reliance of the CUD on anterior interhemispheric pathways, along with the age-related degradation of the CC, could explain the documented elevation of CUD in older adults.

Although a prolonged IHTT is a clear indicator of an age-related disruption in interhemispheric communication, the question remains as to whether higher cognitive processes reliant on such communication are also disrupted in old age. One study has recently examined this question by assessing the ability of one hemisphere to inhibit information from the other. Delvenne and Castronovo (2018) used a divided-field Stroop paradigm wherein the target and distracter were spatially separated and presented either both within a single hemifield or each in a different hemifield. The study revealed that young adults exhibited consistent Stroop interference levels, irrespective of whether the target and distracter were presented unilaterally or bilaterally. This implies that a fully developed and healthy CC inherently facilitates information exchange between hemispheres, even though such transmission may negatively impact task performance. Conversely, older participants exhibited a noteworthy reduction in interhemispheric interference compared to intrahemispheric interference, suggesting a partial disruption in the automatic nature of interhemispheric transfer. Therefore, emerging research suggests that interhemispheric communication undergoes disruption in old age, potentially influencing higher cognitive processes, including inhibition processing. However, additional research is clearly required to explore further the functional implications of agerelated CC degeneration across various cognitive domains.

In the present study, we investigated potential age-related differences in interhemispheric communication in the context of visual working memory (VWM), the ability to temporarily process and maintain visual information. Similarly to the contralateral organization of visual processing, research has shown that the representations in VWM are, at least to some extent, stored in the hemisphere contralateral to the hemifield in which the information is displayed (Delvenne, 2012; Delvenne et al., 2011; Gratton, 1998; Gratton et al., 1997; Vogel & Machizawa, 2004). The memory traces are then integrated and exchanged across the two hemispheres through the CC (Brincat et al., 2021; Gazzaniga, 2000). In the event of a partial disruption in interhemispheric communication among older individuals, the visual information processed and temporarily stored by one hemisphere might not be exchanged as efficiently with the other hemisphere, unlike in younger adults. In simpler terms, there is a reasonable assumption that older adults could exhibit a more pronounced decline in VWM when the memory and test arrays are presented in opposite hemifields.

To the best of my knowledge, the ability to integrate memory traces across hemifields/hemispheres in older adults has not yet been investigated. In the present study, two experiments using an adaptation of the well-known N-back paradigm were carried out. In a typical N-back task, participants are presented with a sequence of individual stimuli and must indicate whether the stimulus presented matches the stimulus presented N items before. This task was originally developed by Kirchner (1958) and has been extensively used as a measure of working memory. Here, we implemented a hemifield manipulation in the visual one-back task, which involved the stimuli being presented in either the left or right hemifield. Consequently, the two matching stimuli could appear in the same hemifield (e.g., both in the left hemifield) or in opposite hemifields (e.g., the first stimulus in the left hemifield and the second in the right hemifield). When presented in opposite hemifields, effective matching requires interhemispheric communication, as these VWM traces are encoded and stored separately across the left and right hemispheres (e.g., Gratton et al., 1997). If there is a reduction in VWM interhemispheric transfer in older adults, a more pronounced decline in their VWM performance may be observed when the matching stimuli are presented in opposite hemifields compared to when they are in the same hemifield.

Transparency and Openness

The experimental protocol was approved by the School of Psychology Research Ethics Committee, from the University of Leeds (title of the project: "The Effect of Ageing on Visual Working Memory Within and Across Hemifields"; Approval Reference Number: PSYC-126). Participants were treated in accordance with the Declaration of Helsinki. The study design, hypotheses, and analysis plan were not preregistered. The Method section reports how the sample size was determined, any data exclusions, all manipulations, and all measures. De-identified data and materials are available on the Open Science Framework at https://osf.io/5qfpu/?view_only=9d026bf5daaa41a48739cd525fb890cb. Analytic code was not provided for findings that can straightforwardly be reproduced using analyses of variance (ANOVAs). Data were analyzed using SPSS Statistics (Version 29).

Experiment 1

Method

Participants

Based on an a priori power calculation with G^* Power (Version 3.1.9), a total sample size of 54 participants was recommended. This

is to detect a small effect (f=0.25) for a repeated measures ANOVA with within-between interaction between the two groups (young and older adults) and measurements (same and different hemifields) and would result in an actual power of 0.950. According to the sample size calculations, it was recommended to recruit a minimum of 27 participants per age group; however, due to the study being carried out online, a higher sample size was aimed for.

We tested 38 young adults (32 females) aged 18-22 ($M_{age} = 18.74$; SD = 0.76) and 38 older adults (16 females) aged 65–85 ($M_{age} =$ 68.92; SD = 4.06). The racial distributions of the samples were not recorded, and all participants resided in the United Kingdom. Young participants were recruited using the Participant Pool Scheme from the University of Leeds Psychology Department and the online recruitment platform Prolific (https://www.prolific.co). Those recruited from the Participant Pool Scheme received course credits, and those recruited through Prolific received £5 for participating. Older participants were recruited using Prolific and were screened for possible underlying neurological disturbances with the Mini Montreal Cognitive Assessment (Mini MoCA; Nasreddine et al., 2005). The MoCA has been found to have high construct validity (Freitas et al., 2011) and higher diagnostic reliability compared to a similar well-known cognitive impairment screening test, the minimental state examination (Folstein et al., 1975). The recommended cutoff score of 11 was applied, and all the older adults recruited here achieved at least that score. All participants were monolingual (self-reported) and right-handed, as determined by a score of at least +50 on an online version of the Edinburgh Handedness questionnaire (Oldfield, 1971). They also self-reported no history of neurological problems, correct color vision, and normal or corrected-to-normal visual acuity.

Apparatus and Stimuli

The stimuli were displayed on the participants' computer screens, and the experiment was run using the online research platform Gorilla (https://www.gorilla.sc). The stimuli were distorted versions of the revised set of Snodgrass and Vanderwart's (1980) object databank (Rossion & Pourtois, 2004) and appeared abstract and meaningless (see McKeown et al., 2014 and Figure 1). The size of

Figure 1 Examples of the Stimuli Used in This Study



each stimulus was 150×150 pixels. The stimuli were presented laterally at the center of the left or right half of the computer screen against a white background.

Design and Procedure

Data collection started in January 2020 and was completed in April 2020. Participants were provided with a URL link to initiate the experiment. No constraints were implemented to require the usage of specific equipment, except that the experiment could only be conducted on a desktop computer or laptop and was not compatible with tablets or phones. Instructions were shown on the screen, followed up by a consent form that participants had to tick to indicate they agreed to partake in the study. Participants then completed the Edinburgh Handedness questionnaire, the mini MoCA, and finally, the main experimental one-back task. The study employed a mixed two-way (2×2) design with group (young adults vs. older adults) acting as the between-subjects independent variable and display (same-hemifield vs. different-hemifield) as the within-subjects independent variable. Although the main dependent variable was the accuracy score calculated as the percentage of correct responses, reaction times were also recorded. As shown in Figure 2, each trial of the main experimental task started with the presentation of a central black fixation cross at the center of the screen for 250 ms. Participants were encouraged, through instructions, to fixate on the center of the screen. One stimulus was then presented, either on the left or right side of the screen, for 180 ms, a presentation time commonly considered short enough to prevent saccadic eye movements (Bourne, 2006; Carpenter, 1988). From the occurrence of a second stimulus onward, participants were instructed to decide as quickly as possible whether the stimulus matched the one that immediately preceded it. Responses had to be made within 3,000 ms of the onset of the stimulus by pressing the appropriate key on the computer keyboard ("k" = same; "d" = different). The next trial began 750 ms after a response had been made or after 3,750 ms if no response had been recorded.

Out of the 260 stimuli available, 204 stimuli were randomly selected and never repeated throughout the study. The experiment consisted of a training block of 12 trials and six experimental blocks of 32 trials each. Within each block of experimental trials, a stimulus matched the immediately preceding one on eight occasions, meaning that within each block, there were eight matched trials and 24 nonmatched trials. Matched and nonmatched trials were randomly presented within each block. Including fewer matched trials compared to nonmatched trials may not only decrease the probability of guessing but could also incentivize participants to sustain focused attention on the task, as the omission of a matched trial would carry a higher cost. The location (i.e., left or right) of the stimuli on the screen was also randomly displayed with the constraint that half of the matched trials shared the same hemifield (i.e., same-hemifield condition) and the other half were displayed in opposite hemifields (i.e., differenthemifield condition). This resulted in a total of 48 matched trials per participant, with 24 trials in the same-hemifield condition and 24 trials in the different-hemifield condition, and 144 unmatched trials, with 72 trials in the same-hemifield condition and 72 trials in the differenthemifield condition. The complete task, including the mini MoCA, handedness questionnaire, etc., lasted approximately 40 min.



Figure 2 Illustrations of the Different Conditions Used in Experiment 1

Note. (A) Example of a matched trial in the same-hemifield condition. (B) Example of a matched trial in the different-hemifield condition. The stimuli are not drawn to scale for illustration purposes. The symbol "+" represents "cross" in the figure.

Results

Even though accuracy was the primary dependent measure as the study was conducted online and there was no explicit request for speedy responses, mixed-design ANOVAs on RTs, with display (same-hemifield vs. different hemifield) as the within-subjects variable, and group (young vs. older adults) as the between-subjects variable, were first conducted. The mixed-design ANOVA on correctly matched trials revealed no significant effects. The mixed-design ANOVA on correct unmatched trials revealed a marginal effect of group, F(1, 74) = 3.86, p = .053, partial $\eta^2 = .050$, with older adults (M = 939 ms, SD = 229) being slightly slower than young adults (M = 845 ms, SD = 202). There was a significant effect of Display, F(1, 74) = 10.74, p = .002, partial $\eta^2 = .127$, with slower responses in the different-hemifield condition (M = 870 ms, SD = 247) as compared to the same-hemifield condition (M = 870 ms, SD = 189). No other effects were found.

The mixed-design ANOVA on the percentage of correct responses for the matched trials revealed a significant effect of group, F(1, 74) =9.33, p = .003, partial $\eta^2 = .112$, with higher accuracy in young adults (M = 81.62, SD = 16.49) as compared to older adults (M = 71.27, SD = 20.96). The effect of Display was not significant (p = .153), suggesting similar overall performance in the same-hemifield and different-hemifield conditions. Interestingly, as it can be seen clearly in Figure 3A, the Display × Group interaction was significant, F(1,74) = 6.57, p = .012, partial $\eta^2 = .082$, showing an effect of Display only for older adults, t(37) = 2.56, p = .015, but not for young adults (p = .374). The effect of Display in older adults revealed lower accuracy in the different-hemifield condition (M = 66.08, SD = 18.88) as compared to the same-hemifield condition (M = 76.46, SD = 21.88). The Display × Group interaction also showed a significant group effect in the different-hemifield condition, t(74) = 4.15, p < .001, where young adults (M = 83.07, SD = 16.74) performed better than older adults (M = 66.09, SD = 18.88). In contrast, no group effect was found in the same-hemifield condition (p = .405).

The mixed-design ANOVA on the percentage of correct responses for the unmatched trials (Figure 3B) revealed a significant effect of group, F(1, 74) = 7.84, p = .007, partial $\eta^2 = .096$, with higher accuracy in young adults (M = 93.28, SD = 9.61) as compared to older adults (M = 84.86, SD = 17.34). No other effects were found.

Finally, analyses were conducted on the measure of sensitivity A'^1 (Figure 3C). The mixed-design ANOVA on A' revealed a significant effect of group, F(1, 74) = 9.81, p = .002, partial $\eta^2 = .117$, with higher A' values in young adults (M = 0.93, SD = 0.08) as compared to older adults (M = 0.85, SD = 0.15). No other effects were found.

¹ A' increases from 0.5 for chance performance to 1.0 for perfect performance (Macmillan & Creelman, 2005). A' was calculated following the formula developed by Grier (1971): A' = 0.5 + [(H - g)(1 + H - g)/4H(1 - g)], where H is the rate of correct detection (Hit Rate—namely the matched trials), and g is the rate of incorrect detection (guessing rate). When g was greater than H, the following formula was used (Aaronson & Watts, 1987): A' = 0.5 - [(g - H)(1 + g - H)/4g(1 - H)].



Note. Error bars represent standard errors of the mean values.

In summary, the findings indicated a significant decline in VWM performance among older adults compared to younger individuals. This decline was evident across both matched and unmatched trials, as well as in the measure of sensitivity A'. Furthermore, older adults exhibited a notable reduction in VWM performance when matching stimuli were presented in opposite hemifields compared to the same hemifield, suggesting a potential disruption in interhemispheric communication. However, this effect was not found when sensitivity measures were considered, which hinders definitive conclusions at this point.

Experiment 2

In the first experiment, when matching stimuli were presented in different hemifields, they naturally occupied distinct locations. In contrast, in the same-hemifield condition, the matching stimuli shared the exact same location. This raises the possibility that older adults may be more influenced by a change in location between the memory and test stimuli rather than solely by a change of hemifield. This could be particularly significant if older adults encounter difficulties in shifting attention between different locations or in transferring information from iconic memory, the brief sensory memory system that holds visual information for a fraction of a second, to working memory. When the two matching stimuli are presented at the same location, performance might rely on residuals of iconic memory without necessitating the transfer of these sensory representations into VWM. Despite the time between two matching stimuli (i.e., approximately 1950 ms, which includes an average response time of 954 ms in older adults, a fixed delay of 750 ms, and a 250 ms fixation cross) surpassing the duration of iconic memory (Sperling, 1960), and the likelihood of the fixation cross acting as a visual mask that disrupts iconic memory, a second experiment was conducted to explicitly eliminate this possibility.

Method

Participants

Because this second experiment used the same number of trials and variables as in the first experiment, the same a priori power calculation was used with a recommended total sample size of at least 54 participants. We tested 37 new young adults (25 females) aged 18–28 ($M_{age} = 21.19$; SD = 2.51) and 37 new older adults (19 females) aged 65–74 ($M_{age} = 68.32$; SD = 2.73). The racial distributions of the samples were not recorded, and all participants resided in the United Kingdom. Participants were recruited and screened in the same way as for Experiment 1. All older adults passed the mini MoCA, and all participants self-reported being monolingual, having correct color vision and normal or correctedto-normal visual acuity, and no history of neurological problems. They were also right-handed, as determined by a score of at least +50 on the Edinburgh Handedness questionnaire.

Stimuli and Procedure

Participants completed the same one-back task as in Experiment 1. However, the following changes were made: the stimuli were presented laterally at both the center of the left half or right half of the computer screen and the center of the top half or bottom half of the screen. As a result, the stimuli were presented at four different locations on the screen: top left, bottom left, top right, or bottom right. The location of the stimuli on the screen was randomly displayed with the constraint that half of the matched trials shared the same hemifield but not the same location (e.g., one stimulus at the bottom-left and the matching stimulus at the top-left) and the other half of the matched trials were displayed in opposite hemifields (e.g., one stimulus at the bottom-left and the matching stimulus at the bottom-right; see Figure 4). The number of trials and all the other parameters of the experiment were identical to Experiment 1.

Results

Regarding the RTs, the mixed-design ANOVA on the correct matched trials revealed a significant effect of group, F(1, 72) = 4.99, p = .029, partial $\eta^2 = .065$, with older adults (M = 821 ms,

SD = 242) being slower than young adults (M = 715 ms, SD = 167). No other effects were found. The analyses on the correct unmatched trials revealed a significant effect of group, F(1, 72) = 8.05, p = .006, partial $\eta^2 = .101$, with older adults (M = 829 ms, SD = 216) being slower than young adults (M = 700 ms, SD = 176). No other effects were found.

For accuracy, the mixed-design ANOVA on the percentage of correct responses for the matched trials (Figure 5A) revealed a significant effect of group, F(1, 72) = 4.29, p = .042, partial $\eta^2 = .056$, with higher accuracy in young adults (M = 77.7, SD = 13) as compared to older adults (M = 70.8, SD = 17.35). The effect of Display was significant, F(1, 72) = 12.65, p < .001, partial $\eta^2 = .149$, with better accuracy in the same-hemifield condition (M = 76.33,SD = 14.64) than in the different-hemifield condition (M = 72.12, SD = 16.46). Similarly to Experiment 1, the Display \times Group interaction was also significant, F(1, 72) = 4.03, p < .05, partial $\eta^2 = .053$, showing better accuracy in the same-hemifield condition (M = 86.22, SD = 13.87) than in the different-hemifield condition (M = 82.68, SD = 16.19) for older adults, t(36) = 4.00, p < .001, but not for young adults (p = .288). The Display \times Group interaction also showed a significant group effect in the different-hemifield condition, t(72) = 2.53, p = .014, where young adults (M = 76.74, SD = 13.59)

performed better than older adults (M = 67.36, SD = 17.97). In contrast, no group effect was found in the same-hemifield condition (p = .195).

The mixed-design ANOVA on the percentage of correct responses for the unmatched trials (Figure 5B) did not reveal any significant effects.

The analyses on A' (Figure 5C) revealed a significant effect of group, F(1, 72) = 5.45, p = .022, partial $\eta^2 = .070$, with higher A' values in young adults (M = 0.91, SD = 0.07) as compared to older adults (M = 0.84, SD = 0.15). The effect of Display was significant, F(1, 72) = 16.88, p < .001, partial $\eta^2 = .190$, with higher A' values in the same-hemifield condition (M = 0.89, SD = 0.11) than in the different-hemifield condition (M = 0.86, SD = 0.13). The Display × Group interaction was also significant, F(1, 72) = 5.48, p = .022, partial $\eta^2 = .071$, showing a significant difference in A' values between the same-hemifield condition (M = 0.86, SD = 0.14) and the different-hemifield condition (M = 0.83, SD = 0.16) for older adults, t(36) = 3.88, p < .001, but not for young adults (p = .122). The Display \times Group interaction also showed a significant difference in A' values between young adults (M = 0.90, SD = 0.07) and older adults (M = 0.83, SD = 0.16) in the different-hemifield condition, t(51) =2.59, p = .013, but not in the same-hemifield condition (p = .056).

Figure 4

Illustrations of the Different Conditions Used in Experiment 2



Note. (A) Example of a matched trial in the same-hemifield condition. (B) Example of a matched trial in the different-hemifield condition. The stimuli are not drawn to scale for illustration purposes. The symbol "+" represents "cross" in the figure.



Note. Error bars represent standard errors of the mean values.

In summary, the findings from this second experiment further underscored the significant decline in VWM performance observed among older adults compared to their younger counterparts. Compared to younger participants, older adults not only exhibited slower response times for both matched and unmatched trials but also demonstrated lower accuracy in matched trials and reduced scores in sensitivity data. Consistent with Experiment 1, older adults also showed a distinct decline in VWM performance when matching stimuli were presented in opposite hemifields compared to the same hemifield. However, unlike Experiment 1, this effect was also corroborated by lower A' values in the different-hemifield condition compared to the same-hemifield condition among older adults. Additionally, the results from this second experiment suggest that this effect cannot be solely attributed to changes in stimulus location or performance influenced by iconic memory residuals. Instead, they compellingly indicate a decline in interhemispheric communication within VWM associated with ageing.

General Discussion

The primary objective of this study was to examine how typical ageing affects interhemispheric communication in the context of VWM. Previous research has predominantly explored the relationship between age-related CC degeneration and interhemispheric communication, primarily using visuomotor IHTT assessments. However, the specific impact of ageing on higher cognitive processes relying on interhemispheric communication has received less attention (Delvenne & Castronovo, 2018). To investigate this, a divided-field one-back paradigm was used, exploring the influence of ageing on the ability to transfer visual memory traces between hemispheres by presenting visual stimuli in either the left or right hemifield. Given the contralateral organization of visual memories (Delvenne, 2012; Delvenne et al., 2011; Gratton, 1998; Gratton et al., 1997; Vogel & Machizawa, 2004), where the two hemispheres need to communicate for accurate matching of bilateral stimuli, the present study aimed to uncover age-related effects on this process.

The results revealed that young adults maintained consistent VWM performance regardless of whether stimuli appeared in the same or opposite hemifields. This suggests that a fully developed and healthy CC facilitates the exchange of memory traces between hemispheres without additional cost. In contrast, older participants showed a notable decrease in VWM performance when matching stimuli were presented in opposite hemifields compared to the same hemifield, indicating a potential disruption in interhemispheric communication. It is crucial to note that this decline in performance among older adults was not simply due to speed-accuracy tradeoffs, as their reduced performance was not associated with shorter reaction times. This effect was confirmed by sensitivity analysis in the second experiment, revealing lower sensitivity in older adults for between-hemifield trials compared to same-hemifield trials. Although it is not clear why the sensitivity analysis did not confirm the effect in Experiment 1, Experiment 2 provided a clearer picture as it eliminated factors such as a mere change in stimulus location or reliance on iconic memory residuals. This strengthens the argument that the observed decline in VWM performance in this study is, at least partially, linked to disrupted interhemispheric communication associated with ageing.

Given the degenerative process observed in the CC of older adults (Aboitiz et al., 1992; Burke & Yeo, 1994; Davis et al., 2009; Hou & Pakkenberg, 2012; McLaughlin et al., 2007; Michielse et al., 2010; Ota et al., 2006; Sullivan et al., 2002, 2010), it is reasonable to suggest that the reduced interhemispheric transfer of VWM traces may directly result from callosal atrophy associated with typical ageing. This assertion aligns with recent findings from a Diffusion Tensor Imaging study (Qin et al., 2016), which indicated a positive relationship between the integrity of the posterior CC and the occurrence of opposite hemifield illusory conjunctions in a VWM task. Specifically, individuals with greater posterior CC integrity exhibited more bilateral illusory conjunctions, implying stronger interhemispheric transfer. The findings also align with existing literature on age-related differences in interhemispheric transfer relying on the anterior section of the CC, which connects bilateral frontal cortex areas (Gawryluk et al., 2011; Omura et al., 2004; Tettamanti et al., 2002; Weber et al., 2005). Although recent findings have shown that specific tracts of the posterior CC, namely temporal and parietal tracts, have lower matter integrity in typical aging (Delvenne et al., 2021), the age-related CC atrophy seems to follow a gradient of decline from anterior to posterior (Lebel et al., 2012). Specifically, the typical ageing process appears to predominantly impact the small-diameter commissural tracts of the anterior CC section, referred to as the genu, projecting to bilateral areas of the frontal cortex (Bastin et al., 2008; Davis et al., 2009; Fan et al., 2019; Hou & Pakkenberg, 2012; Salat et al., 2005; Schulte et al., 2005; Sullivan et al., 2010). This pattern aligns with age-related cognitive decline observed in functions localized in the frontal regions, such as task switching (Wasylyshyn et al., 2011), declarative memory (Rosen et al., 2002), problem solving (Zahr et al., 2009), processing speed, and working memory (Fling et al., 2011; Madden et al., 2009; Milham et al., 2002; Reuter-Lorenz, 2002). Given that the frontal, and especially the prefrontal, cortex plays a crucial role in VWM (Goldman-Rakic, 2011; Miller, 2000; Miller & Cohen, 2001; Owen et al., 2005), the age-related atrophy of the anterior part of the CC appears to be a plausible explanation for the observed decline in the interhemispheric transfer of VWM traces, affecting both basic visuomotor interhemispheric transfer (Bellis & Wilber, 2001; Davis et al., 2012; Jeeves & Moes, 1996; Reuter-Lorenz & Stanczak, 2000) and higher cognitive processes dependent on such communication (Delvenne & Castronovo, 2018).

However, the cross-sectional design of the study limits the ability to establish firm causation or understand the trajectory of age-related differences. Future longitudinal studies could provide a more comprehensible understanding of CC degeneration and its impact on interhemispheric transfer, including the transfer of VWM. Moreover, the study did not delve into the neural mechanisms underlying the observed effects. A more direct link between the integrity of the CC and the capacity to transfer memory traces between hemispheres would need to be confirmed by neuroimaging studies, using, for instance, diffusion tensor imaging (Beaulieu, 2002). Additionally, investigating gender and handedness as variables would be necessary, given known differences in brain organization among individuals. Indeed, a stronger brain asymmetry is commonly found in male and right-handed individuals, relative to female and lefthanded individuals, respectively, across a range of visual, auditory, and tactile laterality tasks (see the review by Hirnstein et al., 2019).

The current investigation also ought to inspire subsequent research endeavors aimed at examining the relationship between the reduced interhemispheric transfer of simple visual memory traces associated with ageing and the overall age-related decline evident in VWM (Brockmole et al., 2008; Brockmole & Logie, 2013; Brown & Brockmole, 2010; Bruyer & Scailquin, 1999; Chen & Naveh-Benjamin, 2012; Ko et al., 2014; Nicholls & English, 2020; Noack et al., 2012; Peich et al., 2013; Pertzov et al., 2015; Peterson & Naveh-Benjamin, 2016; Rhodes et al., 2017; Tas et al., 2020). The underlying mechanisms for age-related decline in VWM are likely to be multifarious and may include lower perceptual abilities (Faubert & Bellefeuille, 2002; Sara & Faubert, 2000; Schneider & Pichora-Fuller, 2000; Tagliabue et al., 2020), slower processing speed (Gazzaley et al., 2008; Zanto et al., 2010), reduced efficiency of executive attentional capacity (Braver & West, 2011; Gazzaley et al., 2005, 2008; Jost et al., 2011; Reuter-Lorenz & Lustig, 2016; West, 1999), and decreased ability to maintain the stored visual information (Beigneux et al., 2007; Leonards et al., 2002; Palladino & De Beni, 1999). Proposed alterations associated with ageing, encompassing

changes at the neural level such as diminished occipital (Davis et al., 2008; Grady, 1996; Payer et al., 2006; Spreng et al., 2010) and frontal (Milham et al., 2002; Reuter-Lorenz, 2002) cortex activation during perception, attention, and working memory tasks, have been suggested to contribute to the decline in VWM. The present study implies that older adults might exhibit a reduced ability to transfer VWM traces between hemispheres, which could potentially be an additional factor to the overall age-related decline in VWM.

In conclusion, the present findings contribute to the growing body of evidence linking age-related CC degeneration to cognitive decline. The observed decline in VWM performance among older adults, particularly when interhemispheric communication is required, emphasizes the functional relevance of CC integrity for higher cognitive processes. The findings of the present study provide a solid basis for future research into the underlying neural mechanisms underpinning the observed age-related differences. Furthermore, the one-back task employed in this study has theoretically the potential to be used and manipulated to investigate certain clinical conditions characterized by reported abnormalities of the CC, such as Alzheimer's disease and mild cognitive impairment (Di Paola et al., 2010).

References

- Aaronson, D., & Watts, B. (1987). Extensions of Grier's computational formulas for A' and B" to below-chance performance. *Psychological Bulletin*, 102(3), 439–442. https://doi.org/10.1037/0033-2909.102.3.439
- Aboitiz, F., Scheibel, A. B., Fisher, R. S., & Zaidel, E. (1992). Fiber composition of the human corpus callosum. *Brain Research*, 598(1–2), 143–153. https://doi.org/10.1016/0006-8993(92)90178-C
- Bastin, M. E., Piatkowski, J. P., Storkey, A. J., Brown, L. J., Maclullich, A. M. J., & Clayden, J. D. (2008). Tract shape modelling provides evidence of topological change in corpus callosum genu during normal ageing. *NeuroImage*, 43(1), 20–28. https://doi.org/10.1016/j.neuroimage .2008.06.047
- Beaulieu, C. (2002). The basis of anisotropic water diffusion in the nervous system—A technical review. NMR in Biomedicine, 15(7–8), 435–455. https://doi.org/10.1002/nbm.782
- Beigneux, K., Plaie, T., & Isingrini, M. (2007). Aging effect on visual and spatial components of working memory. *The International Journal of Aging & Human Development*, 65(4), 301–314. https://doi.org/10.2190/ AG.65.4.b
- Bellis, T. J., & Wilber, L. A. (2001). Effects of aging and gender on interhemispheric function. *Journal of Speech, Language, and Hearing Research*, 44(2), 246–263. https://doi.org/10.1044/1092-4388(2001/021)
- Bourne, V. J. (2006). The divided visual field paradigm: Methodological considerations. *Laterality: Asymmetries of Body, Brain and Cognition*, 11(4), 373–393. https://doi.org/10.1080/13576500600633982
- Braver, T. S., & West, R. (2011). Working memory, executive control, and aging. In F. I. M. Craik & T. A. Salthouse (Eds.), *The handbook of aging* and cognition: *Third edition* (pp. 311–372). Taylor and Francis. https:// doi.org/10.4324/9780203837665
- Brincat, S. L., Donoghue, J. A., Mahnke, M. K., Kornblith, S., Lundqvist, M., & Miller, E. K. (2021). Interhemispheric transfer of working memories. *Neuron*, 109(6), 1055–1066.e4. https://doi.org/10.1016/j.neuro n.2021.01.016
- Brockmole, J. R., & Logie, R. H. (2013). Age-related change in visual working memory: A study of 55,753 participants aged 8–75. *Frontiers in Psychology*, 4, Article 12. https://doi.org/10.3389/fpsyg.2013.00012
- Brockmole, J. R., Parra, M. A., Sala, S. D., & Logie, R. H. (2008). Do binding deficits account for age-related decline in visual working memory?

Psychonomic Bulletin & Review, 15(3), 543–547. https://doi.org/10.3758/ PBR.15.3.543

- Brown, L. A., & Brockmole, J. R. (2010). The role of attention in binding visual features in working memory: Evidence from cognitive ageing. *The Quarterly Journal of Experimental Psychology: Human Experimental Psychology*, 63(10), 2067–2079. https://doi.org/10.1080/174702110037 21675
- Bruyer, R., & Scailquin, J.-C. (1999). Assessment of visuospatial short-term memory and effect of aging. European Review of Applied Psychology/ Revue Européenne de Psychologie Appliquée, 49(3), 175–180.
- Burke, H. L., & Yeo, R. A. (1994). Systematic variations in callosal morphology: The effects of age, gender, hand preference, and anatomic asymmetry. *Neuropsychology*, 8(4), 563–571. https://doi.org/10.1037/ 0894-4105.8.4.563
- Carpenter, R. H. S. (1988). *Movements of the eyes* (2nd rev. & enlarged ed.). Pion.
- Chen, T., & Naveh-Benjamin, M. (2012). Assessing the associative deficit of older adults in long-term and short-term/working memory. *Psychology* and Aging, 27(3), 666–682. https://doi.org/10.1037/a0026943
- Davis, S. W., Dennis, N. A., Buchler, N. G., White, L. E., Madden, D. J., & Cabeza, R. (2009). Assessing the effects of age on long white matter tracts using diffusion tensor tractography. *NeuroImage*, 46(2), 530–541. https:// doi.org/10.1016/j.neuroimage.2009.01.068
- Davis, S. W., Dennis, N. A., Daselaar, S. M., Fleck, M. S., & Cabeza, R. (2008). Que PASA? The posterior-anterior shift in aging. *Cerebral Cortex*, 18(5), 1201–1209. https://doi.org/10.1093/cercor/bhm155
- Davis, S. W., Kragel, J. E., Madden, D. J., & Cabeza, R. (2012). The architecture of cross-hemispheric communication in the aging brain: Linking behavior to functional and structural connectivity. *Cerebral Cortex*, 22(1), 232–242. https://doi.org/10.1093/cercor/bhr123
- Delvenne, J.-F. (2012). Visual short-term memory and the bilateral field advantage. In G. Kalivas & S. F. Petralia (Eds.), *Short-term memory: New research* (pp. 129–154). Nova Publishers.
- Delvenne, J.-F. (2024). Reduced inter-hemispheric transfer in older adults: Evidence from a divided visual field one-back task. PsyArXiv. https://osf .io/5qfpu/?view_only=9d026bf5daaa41a48739cd525fb890cb
- Delvenne, J.-F., & Castronovo, J. (2018). Reduced inter-hemispheric interference in ageing: Evidence from a divided field Stroop paradigm. *Brain and Cognition*, 122, 26–33. https://doi.org/10.1016/j.bandc.2018 .01.008
- Delvenne, J.-F., Kaddour, L. A., & Castronovo, J. (2011). An electrophysiological measure of visual short-term memory capacity within and across hemifields. *Psychophysiology*, 48(3), 333–336. https://doi.org/10.1111/j .1469-8986.2010.01079.x
- Delvenne, J.-F., Scally, B., Bunce, D., & Burke, M. R. (2021). Splenium tracts of the corpus callosum degrade in old age. *Neuroscience Letters*, 742, Article 135549. https://doi.org/10.1016/j.neulet.2020.135549
- Di Paola, M., Spalletta, G., & Caltagirone, C. (2010). In vivo structural neuroanatomy of corpus callosum in Alzheimer's disease and mild cognitive impairment using different MRI techniques: A review. *Journal of Alzheimer's Disease*, 20(1), 67–95. https://doi.org/10.3233/JAD-2010-1370
- Fan, Q., Tian, Q., Ohringer, N. A., Nummenmaa, A., Witzel, T., Tobyne, S. M., Klawiter, E. C., Mekkaoui, C., Rosen, B. R., Wald, L. L., Salat, D. H., & Huang, S. Y. (2019). Age-related alterations in axonal microstructure in the corpus callosum measured by high-gradient diffusion MRI. *NeuroImage*, 191, 325–336. https://doi.org/10.1016/j.neuroimage .2019.02.036
- Faubert, J., & Bellefeuille, A. (2002). Aging effects on intra- and interattribute spatial frequency information for luminance, color, and working memory. *Vision Research*, 42(3), 369–378. https://doi.org/10.1016/ S0042-6989(01)00292-9
- Fling, B. W., Walsh, C. M., Bangert, A. S., Reuter-Lorenz, P. A., Welsh, R. C., & Seidler, R. D. (2011). Differential callosal contributions to

bimanual control in young and older adults. Journal of Cognitive Neuroscience, 23(9), 2171–2185. https://doi.org/10.1162/jocn.2010.21600

- Folstein, M. F., Folstein, S. E., & McHugh, P. R. (1975). "Mini-mental state." A practical method for grading the cognitive state of patients for the clinician. *Journal of Psychiatric Research*, 12(3), 189–198. https:// doi.org/10.1016/0022-3956(75)90026-6
- Freitas, S., Simões, M. R., Alves, L., & Santana, I. (2011). Montreal Cognitive Assessment (MoCA): Normative study for the Portuguese population. *Journal of Clinical and Experimental Neuropsychology*, 33(9), 989–996. https://doi.org/10.1080/13803395.2011.589374
- Gawryluk, J. R., D'Arcy, R. C. N., Mazerolle, E. L., Brewer, K. D., & Beyea, S. D. (2011). Functional mapping in the corpus callosum: A 4T fMRI study of white matter. *NeuroImage*, 54(1), 10–15. https://doi.org/10.1016/ j.neuroimage.2010.07.028
- Gazzaley, A., Clapp, W., Kelley, J., McEvoy, K., Knight, R. T., & D'Esposito, M. (2008). Age-related top-down suppression deficit in the early stages of cortical visual memory processing. *Proceedings of the National Academy of Sciences of the United States of America*, 105(35), 13122–13126. https://doi.org/10.1073/pnas.0806074105
- Gazzaley, A., Cooney, J. W., Rissman, J., & D'Esposito, M. (2005). Topdown suppression deficit underlies working memory impairment in normal aging. *Nature Neuroscience*, 8(10), 1298–1300. https://doi.org/10 .1038/nn1543
- Gazzaniga, M. S. (2000). Cerebral specialization and interhemispheric communication: Does the corpus callosum enable the human condition? *Brain*, 123(7), 1293–1326. https://doi.org/10.1093/brain/123.7.1293
- Goldman-Rakic, P. S. (2011). Circuitry of primate prefrontal cortex and regulation of behavior by representational memory. In R. Terjung (Ed.), *Comprehensive physiology* (pp. 373–417). Wiley. https://doi.org/10.1002/ cphy.cp010509
- Grady, C. L. (1996). Age-related changes in cortical blood flow activation during perception and memory. *Annals of the New York Academy of Sciences*, 777(1), 14–21. https://doi.org/10.1111/j.1749-6632.1996.tb34 396.x
- Gratton, G. (1998). The contralateral organization of visual memory: A theoretical concept and a research tool. *Psychophysiology*, *35*(6), 638–647. https://doi.org/10.1111/1469-8986.3560638
- Gratton, G., Corballis, P. M., & Jain, S. (1997). Hemispheric organization of visual memories. *Journal of Cognitive Neuroscience*, 9(1), 92–104. https://doi.org/10.1162/jocn.1997.9.1.92
- Grier, J. B. (1971). Nonparametric indexes for sensitivity and bias: Computing formulas. *Psychological Bulletin*, 75(6), 424–429. https:// doi.org/10.1037/h0031246
- Hirnstein, M., Hugdahl, K., & Hausmann, M. (2019). Cognitive sex differences and hemispheric asymmetry: A critical review of 40 years of research. *Laterality*, 24(2), 204–225. https://doi.org/10.1080/1357650X .2018.1497044
- Hou, J., & Pakkenberg, B. (2012). Age-related degeneration of corpus callosum in the 90+ years measured with stereology. *Neurobiology of Aging*, 33(5), 1009.e1–1009.e9. https://doi.org/10.1016/j.neurobiolaging .2011.10.017
- Jeeves, M. A., & Moes, P. (1996). Interhemispheric transfer time differences related to aging and gender. *Neuropsychologia*, 34(7), 627–636. https:// doi.org/10.1016/0028-3932(95)00157-3
- Jokinen, H., Ryberg, C., Kalska, H., Ylikoski, R., Rostrup, E., Stegmann, M. B., Waldemar, G., Madureira, S., Ferro, J. M., van Straaten, E. C. W., Scheltens, P., Barkhof, F., Fazekas, F., Schmidt, R., Carlucci, G., Pantoni, L., Inzitari, D., Erkinjuntti, T., & the LADIS group. (2007). Corpus callosum atrophy is associated with mental slowing and executive deficits in subjects with age-related white matter hyperintensities: The LADIS Study. *Journal of Neurology, Neurosurgery, and Psychiatry*, 78(5), 491–496. https://doi.org/10.1136/jnnp .2006.096792

- Jost, K., Bryck, R. L., Vogel, E. K., & Mayr, U. (2011). Are old adults just like low working memory young adults? Filtering efficiency and age differences in visual working memory. *Cerebral Cortex*, 21(5), 1147–1154. https://doi.org/10.1093/cercor/bhq185
- Kirchner, W. K. (1958). Age differences in short-term retention of rapidly changing information. *Journal of Experimental Psychology*, 55(4), 352– 358. https://doi.org/10.1037/h0043688
- Ko, P. C., Duda, B., Hussey, E., Mason, E., Molitor, R. J., Woodman, G. F., & Ally, B. A. (2014). Understanding age-related reductions in visual working memory capacity: Examining the stages of change detection. *Attention, Perception, & Psychophysics*, 76(7), 2015–2030. https:// doi.org/10.3758/s13414-013-0585-z
- Lebel, C., Gee, M., Camicioli, R., Wieler, M., Martin, W., & Beaulieu, C. (2012). Diffusion tensor imaging of white matter tract evolution over the lifespan. *NeuroImage*, 60(1), 340–352. https://doi.org/10.1016/j.neuroima ge.2011.11.094
- Leonards, U., Ibanez, V., & Giannakopoulos, P. (2002). The role of stimulus type in age-related changes of visual working memory. *Experimental Brain Research*, 146(2), 172–183. https://doi.org/10.1007/s00221-002-1175-9
- Linnet, E., & Roser, M. E. (2012). Age-related differences in interhemispheric visuomotor integration measured by the redundant target effect. *Psychology and Aging*, 27(2), 399–409. https://doi.org/10.1037/a0024905
- Macmillan, N. A., & Creelman, C. D. (2005). Detection theory: A user's guide (2nd ed.). Lawrence Erlbaum. https://doi.org/10.4324/97814 10611147
- Madden, D. J., Spaniol, J., Costello, M. C., Bucur, B., White, L. E., Cabeza, R., Davis, S. W., Dennis, N. A., Provenzale, J. M., & Huettel, S. A. (2009). Cerebral white matter integrity mediates adult age differences in cognitive performance. *Journal of Cognitive Neuroscience*, 21(2), 289–302. https:// doi.org/10.1162/jocn.2009.21047
- McKeown, D., Holt, J., Delvenne, J.-F., Smith, A., & Griffiths, B. (2014). Active versus passive maintenance of visual nonverbal memory. *Psychonomic Bulletin & Review*, 21(4), 1041–1047. https://doi.org/10 .3758/s13423-013-0574-1
- McLaughlin, N. C. R., Paul, R. H., Grieve, S. M., Williams, L. M., Laidlaw, D., DiCarlo, M., Clark, C. R., Whelihan, W., Cohen, R. A., Whitford, T. J., & Gordon, E. (2007). Diffusion tensor imaging of the corpus callosum: A cross-sectional study across the lifespan. *International Journal of Developmental Neuroscience*, 25(4), 215–221. https://doi.org/10.1016/j .ijdevneu.2007.03.008
- Michielse, S., Coupland, N., Camicioli, R., Carter, R., Seres, P., Sabino, J., & Malykhin, N. (2010). Selective effects of aging on brain white matter microstructure: A diffusion tensor imaging tractography study. *NeuroImage*, 52(4), 1190–1201. https://doi.org/10.1016/j.neuroimage.2010.05.019
- Milham, M. P., Erickson, K. I., Banich, M. T., Kramer, A. F., Webb, A., Wszalek, T., & Cohen, N. J. (2002). Attentional control in the aging brain: Insights from an fMRI study of the stroop task. *Brain and Cognition*, 49(3), 277–296. https://doi.org/10.1006/brcg.2001.1501
- Miller, E. K. (2000). The prefrontal cortex and cognitive control. Nature Reviews Neuroscience, 1(1), 59–65. https://doi.org/10.1038/35036228
- Miller, E. K., & Cohen, J. D. (2001). An integrative theory of prefrontal cortex function. *Annual Review of Neuroscience*, 24, 167–202. https:// doi.org/10.1146/annurev.neuro.24.1.167
- Murman, D. L. (2015). The impact of age on cognition. *Seminars in Hearing*, 36(3), 111–121. https://doi.org/10.1055/s-0035-1555115
- Nasreddine, Z. S., Phillips, N. A., Bédirian, V., Charbonneau, S., Whitehead, V., Collin, I., Cummings, J. L., & Chertkow, H. (2005). The Montreal Cognitive Assessment, MoCA: A brief screening tool for mild cognitive impairment. *Journal of the American Geriatrics Society*, 53(4), 695–699. https://doi.org/10.1111/j.1532-5415.2005.53221.x
- Nicholls, L. A. B., & English, B. (2020). Multimodal coding and strategic approach in young and older adults' visual working memory performance. *Aging, Neuropsychology, and Cognition*, 27(1), 83–113. https://doi.org/ 10.1080/13825585.2019.1585515

- Noack, H., Lövdén, M., & Lindenberger, U. (2012). Normal aging increases discriminal dispersion in visuospatial short-term memory. *Psychology and Aging*, 27(3), 627–637. https://doi.org/10.1037/a0027251
- Oldfield, R. C. (1971). The assessment and analysis of handedness: The Edinburgh inventory. *Neuropsychologia*, 9(1), 97–113. https://doi.org/10 .1016/0028-3932(71)90067-4
- Omura, K., Tsukamoto, T., Kotani, Y., Ohgami, Y., Minami, M., & Inoue, Y. (2004). Different mechanisms involved in interhemispheric transfer of visuomotor information. *Neuroreport*, 15(18), 2707–2711.
- Ota, M., Obata, T., Akine, Y., Ito, H., Ikehira, H., Asada, T., & Suhara, T. (2006). Age-related degeneration of corpus callosum measured with diffusion tensor imaging. *NeuroImage*, 31(4), 1445–1452. https://doi.org/ 10.1016/j.neuroimage.2006.02.008
- Owen, A. M., McMillan, K. M., Laird, A. R., & Bullmore, E. (2005). N-back working memory paradigm: A meta-analysis of normative functional neuroimaging studies. *Human Brain Mapping*, 25(1), 46–59. https:// doi.org/10.1002/hbm.20131
- Palladino, P., & De Beni, R. (1999). Working memory in aging: Maintenance and suppression. Aging, 11(5), 301–306. https://doi.org/10.1007/BF03339804
- Payer, D., Marshuetz, C., Sutton, B., Hebrank, A., Welsh, R. C., & Park, D. C. (2006). Decreased neural specialization in old adults on a working memory task. *Neuroreport*, *17*(5), 487–491. https://doi.org/10.1097/01 .wnr.0000209005.40481.31
- Peich, M. C., Husain, M., & Bays, P. M. (2013). Age-related decline of precision and binding in visual working memory. *Psychology and Aging*, 28(3), 729–743. https://doi.org/10.1037/a0033236
- Pertzov, Y., Heider, M., Liang, Y., & Husain, M. (2015). Effects of healthy ageing on precision and binding of object location in visual short term memory. *Psychology and Aging*, 30(1), 26–35. https://doi.org/10.1037/ a0038396
- Peterson, D. J., & Naveh-Benjamin, M. (2016). The role of aging in intraitem and item-context binding processes in visual working memory. *Journal of Experimental Psychology: Learning, Memory, and Cognition*, 42(11), 1713–1730. https://doi.org/10.1037/xlm0000275
- Poffenberger, A. T. (1912). Reaction time to retinal stimulation: With special reference to the time lost in conduction through nerve centers. Science Press.
- Qin, S., Ray, N. R., Ramakrishnan, N., Nashiro, K., O'Connell, M. A., & Basak, C. (2016). Illusory conjunctions in visual short-term memory: Individual differences in corpus callosum connectivity and splitting attention between the two hemifields. *Psychophysiology*, 53(11), 1639– 1650. https://doi.org/10.1111/psyp.12735
- Reuter-Lorenz, P. A. (2002). New visions of the aging mind and brain. Trends in Cognitive Sciences, 6(9), P394–P400. https://doi.org/10.1016/ S1364-6613(02)01957-5
- Reuter-Lorenz, P. A., & Lustig, C. (2016). Working memory and executive functions in the aging brain. In R. Cabeza, L. Nyberg, & D. C. Park (Eds.), *Cognitive neuroscience of aging: Linking cognitive and cerebral aging* (2nd ed., pp. 235–258). Oxford University Press. https://doi.org/10.1093/ acprof:oso/9780199372935.003.0010
- Reuter-Lorenz, P. A., & Stanczak, L. (2000). Differential effects of aging on the functions of the corpus callosum. *Developmental Neuropsychology*, 18(1), 113–137. https://doi.org/10.1207/S15326942DN1801_7
- Rhodes, S., Parra, M. A., Cowan, N., & Logie, R. H. (2017). Healthy aging and visual working memory: The effect of mixing feature and conjunction changes. *Psychology and Aging*, 32(4), 354–366. https://doi.org/10.1037/ pag0000152
- Rosen, A. C., Prull, M. W., O'Hara, R., Race, E. A., Desmond, J. E., Glover, G. H., Yesavage, J. A., & Gabrieli, J. D. E. (2002). Variable effects of aging on frontal lobe contributions to memory. *Neuroreport*, *13*(18), 2425–2428. https://doi.org/10.1097/00001756-200212200-00010
- Rossion, B., & Pourtois, G. (2004). Revisiting Snodgrass and Vanderwart's object pictorial set: The role of surface detail in basic-level object recognition. *Perception*, 33(2), 217–236. https://doi.org/10.1068/p5117

- Ryberg, C., Rostrup, E., Paulson, O. B., Barkhof, F., Scheltens, P., van Straaten, E. C. W., van der Flier, W. M., Fazekas, F., Schmidt, R., Ferro, J. M., Baezner, H., Erkinjuntti, T., Jokinen, H., Wahlund, L. O., Poggesi, A., Pantoni, L., Inzitari, D., Waldemar, G., & the LADIS study group. (2011). Corpus callosum atrophy as a predictor of age-related cognitive and motor impairment: A 3-year follow-up of the LADIS study cohort. *Journal of the Neurological Sciences*, 307(1–2), 100–105. https://doi.org/ 10.1016/j.jns.2011.05.002
- Salat, D. H., Tuch, D. S., Greve, D. N., van der Kouwe, A. J. W., Hevelone, N. D., Zaleta, A. K., Rosen, B. R., Fischl, B., Corkin, S., Rosas, H. D., & Dale, A. M. (2005). Age-related alterations in white matter microstructure measured by diffusion tensor imaging. *Neurobiology of Aging*, 26(8), 1215–1227. https://doi.org/10.1016/j.neurobiolaging.2004.09.017
- Sara, M., & Faubert, J. (2000). Aging, perception, and visual short-term memory for luminance-defined form. *Ophthalmic & Physiological Optics*, 20(4), 314–322. https://doi.org/10.1046/j.1475-1313.2000.00531.x
- Schneider, B. A., & Pichora-Fuller, M. K. (2000). Implications of perceptual deterioration for cognitive aging research. In F. I. M. Craik & T. A. Salthouse (Eds.), *The handbook of aging and cognition* (2nd ed., pp. 155–219). Lawrence Erlbaum.
- Schulte, T., Sullivan, E. V., Müller-Oehring, E. M., Adalsteinsson, E., & Pfefferbaum, A. (2005). Corpus callosal microstructural integrity influences interhemispheric processing: A diffusion tensor imaging study. *Cerebral Cortex*, 15(9), 1384–1392. https://doi.org/10.1093/cercor/bhi020
- Snodgrass, J. G., & Vanderwart, M. (1980). A standardized set of 260 pictures: Norms for name agreement, image agreement, familiarity, and visual complexity. *Journal of Experimental Psychology: Human Learning* and Memory, 6(2), 174–215. https://doi.org/10.1037//0278-7393.6.2.174
- Sperling, G. (1960). The information available in brief visual presentations. Psychological Monographs: General and Applied, 74(11), 1–29. https:// doi.org/10.1037/h0093759
- Spreng, R. N., Wojtowicz, M., & Grady, C. L. (2010). Reliable differences in brain activity between young and old adults: A quantitative meta-analysis across multiple cognitive domains. *Neuroscience and Biobehavioral Reviews*, 34(8), 1178–1194. https://doi.org/10.1016/j.neubiorev.2010 .01.009
- Sullivan, E. V., Pfefferbaum, A., Adalsteinsson, E., Swan, G. E., & Carmelli, D. (2002). Differential rates of regional brain change in callosal and ventricular size: A 4-year longitudinal MRI study of elderly men. *Cerebral Cortex*, 12(4), 438–445. https://doi.org/10.1093/cercor/12 .4.438

- Sullivan, E. V., Rohlfing, T., & Pfefferbaum, A. (2010). Quantitative fiber tracking of lateral and interhemispheric white matter systems in normal aging: Relations to timed performance. *Neurobiology of Aging*, 31(3), 464–481. https://doi.org/10.1016/j.neurobiolaging.2008.04.007
- Tagliabue, C. F., Lombardi, L., & Mazza, V. (2020). Individuation of object parts in aging. Attention, Perception, & Psychophysics, 82(5), 2703–2713. https://doi.org/10.3758/s13414-020-01996-2
- Tas, A. C., Costello, M. C., & Buss, A. T. (2020). Age-related decline in visual working memory: The effect of nontarget objects during a delayed estimation task. *Psychology and Aging*, 35(4), 565–577. https://doi.org/10 .1037/pag0000450
- Tettamanti, M., Paulesu, E., Scifo, P., Maravita, A., Fazio, F., Perani, D., & Marzi, C. A. (2002). Interhemispheric transmission of visuomotor information in humans: fMRI evidence. *Journal of Neurophysiology*, 88(2), 1051–1058. https://doi.org/10.1152/jn.2002.88.2.1051
- Vogel, E. K., & Machizawa, M. G. (2004). Neural activity predicts individual differences in visual working memory capacity. *Nature*, 428(6984), 748–751. https://doi.org/10.1038/nature02447
- Wasylyshyn, C., Verhaeghen, P., & Sliwinski, M. J. (2011). Aging and task switching: A meta-analysis. *Psychology and Aging*, 26(1), 15–20. https:// doi.org/10.1037/a0020912
- Weber, B., Treyer, V., Oberholzer, N., Jaermann, T., Boesiger, P., Brugger, P., Regard, M., Buck, A., Savazzi, S., & Marzi, C. A. (2005). Attention and interhemispheric transfer: A behavioral and fMRI study. *Journal of Cognitive Neuroscience*, 17(1), 113–123. https://doi.org/10.1162/08989 29052880002
- West, R. (1999). Visual distraction, working memory, and aging. *Memory & Cognition*, 27(6), 1064–1072. https://doi.org/10.3758/BF03201235
- Zahr, N. M., Rohlfing, T., Pfefferbaum, A., & Sullivan, E. V. (2009). Problem solving, working memory, and motor correlates of association and commissural fiber bundles in normal aging: A quantitative fiber tracking study. *NeuroImage*, 44(3), 1050–1062. https://doi.org/10.1016/j .neuroimage.2008.09.046
- Zanto, T. P., Toy, B., & Gazzaley, A. (2010). Delays in neural processing during working memory encoding in normal aging. *Neuropsychologia*, 48(1), 13–25. https://doi.org/10.1016/j.neuropsychologia.2009.08.003

Received August 9, 2023

Revision received May 13, 2024

Accepted May 19, 2024